Author's response to reviews

Title: Family-based intervention in adolescent restrictive eating disorders: Early treatment response and low weight suppression is associated with favourable one-year outcome

Authors:

Ingemar Swenne (ingemar.swenne@kbh.uu.se)
Thomas Parling (thomas.parling@psyk.uu.se)
Helena Salonen Ros (helena.salaonen.ros@akademiska.se)

Version: 4 Date: 24 Aug 2017

Author’s response to reviews:

The heading "Declarations" is now added before the reference section.

The ethics approval sections has been amended.

The use of the headings "Funding" and "Acknowledgements" has been corrected.

"Abbreviations" has now been added after the conclusions section.

Email addresses of all authors have been added to the title page,